Author's response to reviews

Title: Externalizing behavior in early childhood and body mass index from age 2 to 12 years: longitudinal analyses of a prospective cohort study

Authors:

Sarah E Anderson (sanderson@cph.osu.edu)
Xin He (xinhe@umd.edu)
Sarah Schoppe-Sullivan (schoppe-sullivan.1@osu.edu)
Aviva Must (aviva.must@tufts.edu)

Version: 4 Date: 8 July 2010

Author's response to reviews:

Removed unnecessary within paragraph bolding. Minor typographical corrections.